%%% PLANTILLA DISEÑADA PARA LA REALIZACIÓN DE TESIS DE GRADO DEL DOCTORADO EN INGENIERÍA INDUSTRIAL DI3
%% AUTOR DE LA PLANTILLA: RICARDO PALMA

\documentclass[12pt,a4paper,oneside]{book}

\usepackage[T1]{fontenc}
\usepackage{times}
\usepackage[utf8]{inputenc}
\usepackage{amsmath}
\usepackage{graphicx}
\usepackage{multicol}
\usepackage{longtable}
\usepackage[refpages]{gloss}
\usepackage{float}
\usepackage{anysize}
\usepackage{appendix}
\usepackage{lscape} 
\usepackage{pdflscape}
\usepackage{multirow}
\usepackage{listings}
\usepackage{color}
\usepackage{setspace}
\usepackage{enumerate} 
\usepackage{pdfpages}
\usepackage{lipsum}


\begin{document}

%----------------------------------------------------------------------------------------
%	CONFIGURACION
%----------------------------------------------------------------------------------------

\marginsize{3.0cm}{3.0cm}{4.0cm}{3.0cm}
\renewcommand*{\contentsname}{ÍNDICE}
\renewcommand*{\listtablename}{Índice de tablas}
\renewcommand*{\listfigurename}{Índice de figuras}
\renewcommand{\baselinestretch}{1.0}
\renewcommand{\appendixname}{Anexos}
\renewcommand{\appendixtocname}{Anexos}
\renewcommand{\appendixpagename}{Anexos}
\renewcommand{\thetable}{\arabic{chapter}.\arabic{table}}
\renewcommand*{\tablename}{Tabla}
\renewcommand*{\chaptername}{Capítulo}
\renewcommand*{\thechapter}{\Roman{chapter}}
\renewcommand{\thesection}{\arabic{chapter}.\arabic{section}}
\renewcommand{\figurename}{Figura}
\renewcommand{\thefigure}{\arabic{chapter}.\arabic{figure}}
\renewcommand{\theequation}{\arabic{chapter}.\arabic{equation}}

%----------------------------------------------------------------------------------------
%	PORTADA
%----------------------------------------------------------------------------------------

\begin{titlepage}
 
\begin{center}
 
 {\huge \bf DOCTORADO EN INGENIERÍA}\\[1.2cm]
 {\huge \bf INDUSTRIAL}\\[1.2cm] 
 
{\Large Doctorado Interinstitucional Di3}\\{\Large SECRETARÍA DE POSGRADO -  Facultad de Ingeniería - Univarsidad Nacional de Jujuy}\\[2.0cm]


\begin{center}
\includegraphics[width=0.45\textwidth]{imagenes/Di3B}
\end{center}

\vspace{1.7cm}
{{\bf Título: }: }Mi primer Tesis
\title{Mi primer Tesis} % titulo de tu tesis para latex
{\bf \large . }\\[1.7cm] % titulo de tu tesis


{EJEMPLAR DE TESIS PRESENTADO PARA OPTAR EL TÍTULO \\Doctor en Ingeniería Industrial}\\[0.5cm]
 
{{\bf Autor}: }\\[0.5cm] % nombres del autor o autores

{\large GATES, Guillermo, 2020}\\[0.8cm] % fecha de sustentación
{UNJU - UNAM - UNT - UNLAR - UNCuyo}
\end{center}

\end{titlepage}

\newpage
\newpage
$\ $
\thispagestyle{empty} % para que no se muestre el número en la página

\includepdf{ACTA-TESIS}%se incluye un PDF con firmas o cualquier otro documento



%----------------------------------------------------------------------------------------
%	Dedicatoria
%----------------------------------------------------------------------------------------

\begin{titlepage}

\begin{flushright}
{\large \bf DEDICATORIA}
\\
\textit{}
\\
\textit{} % agregar tu dedicatoria
\end{flushright}
\end{titlepage}

%----------------------------------------------------------------------------------------
%	Agradecimientos
%----------------------------------------------------------------------------------------

\begin{titlepage}

\begin{flushright}
{\large \bf AGRADECIMIENTO}
\\
\textit{}
{} % agregar tu dedicatoria
\end{flushright}
\end{titlepage}

%----------------------------------------------------------------------------------------
%	TABLA DE CONTENIDOS
%---------------------------------------------------------------------------------------

\tableofcontents
\cleardoublepage
\listoftables
\listoffigures 
%\makegloss
\newpage

%----------------------------------------------------------------------------------------
%	Resumen
%----------------------------------------------------------------------------------------

\chapter*{\centering \large RESUMEN EJECUTIVO} % si no queremos que añada la palabra "Capitulo"
\addcontentsline{toc}{section}{RESUMEN} % si queremos que aparezca en el índice
\markboth{RESUMEN}{RESUMEN} % encabezado
\doublespacing
\lipsum{8}

%----------------------------------------------------------------------------------------
%	Abstract
%   en inglés - no debe contener más de 400 apalbras
%----------------------------------------------------------------------------------------

\chapter*{\centering \large ABSTRACT}
\addcontentsline{toc}{section}{ABSTRACT}
\markboth{ABSTRACT}{ABSTRACT}


%----------------------------------------------------------------------------------------
%	INTRODUCCIÓN
%----------------------------------------------------------------------------------------

\chapter*{\centering \large INTRODUCCIÓN} % si no queremos que añada la palabra "Capitulo"
\addcontentsline{toc}{section}{INTRODUCCIÓN} % si queremos que aparezca en el índice
\markboth{INTRODUCCIÓN}{INTRODUCCIÓN} % encabezado

You are strongly encouraged to use \LaTeXe{} for the
preparation of your camera-ready manuscript together with the
corresponding Springer class file \verb+llncs.cls+. Only if you use
\LaTeXe{} can hyperlinks be generated in the online version
of your manuscript.

The \LaTeX{} source of this instruction file for \LaTeX{} users may be
used as a template. This is
located in the ``authors'' subdirectory in
\url{ftp://ftp.springer.de/pub/tex/latex/llncs/latex2e/instruct/} and
entitled \texttt{typeinst.tex}. There is a separate package for Word 
users. Kindly send the final and checked source
and PDF files of your paper to the Contact Volume Editor. This is
usually one of the organizers of the conference. You should make sure
that the \LaTeX{} and the PDF files are identical and correct and that
only one version of your paper is sent. It is not possible to update
files at a later stage. Please note that we do not need the printed
paper.

We would like to draw your attention to the fact that it is not possible
to modify a paper in any way, once it has been published. This applies
to both the printed book and the online version of the publication.
Every detail, including the order of the names of the authors, should
be checked before the paper is sent to the Volume Editors.

\subsection*{Verificación del PDF a entregar}

Kindly assure that the Contact Volume Editor is given the name and email
address of the contact author for your thesis. The Contact Volume Editor
uses these details to compile a list for our production department at
SPS in India. Once the files have been worked upon, SPS sends a copy of
the final pdf of each paper to its contact author. The contact author is
asked to check through the final pdf to make sure that no errors have
crept in during the transfer or preparation of the files. This should
not be seen as an opportunity to update or copyedit the papers, which is
not possible due to time constraints. Only errors introduced during the
preparation of the files will be corrected.

This round of checking takes place about two weeks after the files have
been sent to the Editorial by the Contact Volume Editor, i.e., roughly
seven weeks before the start of the conference for conference
proceedings, or seven weeks before the volume leaves the printer's, for
post-proceedings. If SPS does not receive a reply from a particular
contact author, within the timeframe given, then it is presumed that the
author has found no errors in the paper. The tight publication schedule
of LNCS does not allow SPS to send reminders or search for alternative
email addresses on the Internet.

In some cases, it is the Contact Volume Editor that checks all the final
pdfs. In such cases, the authors are not involved in the checking phase.

\subsection*{Apellidos compuestos o materno, paterno}

If you have more than one surname, please make sure that the Volume Editor
knows how you are to be listed in the author index.

\subsection*{ISBN a cargo del autor de la tesis}

The copyright form may be downloaded from the ``For Authors"
(Information for LNCS Authors) section of the LNCS Website:
\texttt{www.springer.com/lncs}. Please send your signed copyright form
to the Contact Volume Editor, either as a scanned pdf or by fax or by
courier. One author may sign on behalf of all of the other authors of a
particular paper. Digital signatures are acceptable.

\section*{Preparación de la Tesis}

Springer provides you with a complete integrated \LaTeX{} document class
(\texttt{llncs.cls}) for multi-author books such as those in the LNCS
series. Papers not complying with the LNCS style will be reformatted.
This can lead to an increase in the overall number of pages. We would
therefore urge you not to squash your paper.

Please always cancel any superfluous definitions that are
not actually used in your text. If you do not, these may conflict with
the definitions of the macro package, causing changes in the structure
of the text and leading to numerous mistakes in the proofs.

If you wonder what \LaTeX{} is and where it can be obtained, see the
``\textit{LaTeX project site}'' (\url{http://www.latex-project.org})
and especially the webpage ``\textit{How to get it}''
(\url{http://www.latex-project.org/ftp.html}) respectively.

When you use \LaTeX\ together with our document class file,
\texttt{llncs.cls},
your text is typeset automatically in Computer Modern Roman (CM) fonts.
Please do
\emph{not} change the preset fonts. If you have to use fonts other
than the preset fonts, kindly submit these with your files.

Please use the commands \verb+\label+ and \verb+\ref+ for
cross-references and the commands \verb+\bibitem+ and \verb+\cite+ for
references to the bibliography, to enable us to create hyperlinks at
these places.

For preparing your figures electronically and integrating them into
your source file we recommend using the standard \LaTeX{} \verb+graphics+ or
\verb+graphicx+ package. These provide the \verb+\includegraphics+ command.
In general, please refrain from using the \verb+\special+ command.

Remember to submit any further style files and
fonts you have used together with your source files.

\subsubsection*{Títulos y Subtitulos}

Headings should be capitalized
(i.e., nouns, verbs, and all other words
except articles, prepositions, and conjunctions should be set with an
initial capital) and should,
with the exception of the title, be aligned to the left.
Words joined by a hyphen are subject to a special rule. If the first
word can stand alone, the second word should be capitalized.

Here are some examples of headings: ``Criteria to Disprove
Context-Freeness of Collage Language", ``On Correcting the Intrusion of
Tracing Non-deterministic Programs by Software", ``A User-Friendly and
Extendable Data Distribution System", ``Multi-flip Networks:
Parallelizing GenSAT", ``Self-determinations of Man".

\subsubsection*{Lemmas, Propositions, and Theorems.}

The numbers accorded to lemmas, propositions, and theorems, etc. should
appear in consecutive order, starting with Lemma 1, and not, for
example, with Lemma 11.

\subsection*{Figures}

For \LaTeX\ users, we recommend using the \emph{graphics} or \emph{graphicx}
package and the \verb+\includegraphics+ command.

Please check that the lines in line drawings are not
interrupted and are of a constant width. Grids and details within the
figures must be clearly legible and may not be written one on top of
the other. Line drawings should have a resolution of at least 800 dpi
(preferably 1200 dpi). The lettering in figures should have a height of
2~mm (10-point type). Figures should be numbered and should have a
caption which should always be positioned \emph{under} the figures, in
contrast to the caption belonging to a table, which should always appear
\emph{above} the table; this is simply achieved as matter of sequence in
your source.

Please center the figures or your tabular material by using the \verb+\centering+
declaration. Short captions are centered by default between the margins
and typeset in 9-point type (Fig.~\ref{fig:example} shows an example).
The distance between text and figure is preset to be about 8~mm, the
distance between figure and caption about 6~mm.

To ensure that the reproduction of your illustrations is of a reasonable
quality, we advise against the use of shading. The contrast should be as
pronounced as possible.

If screenshots are necessary, please make sure that you are happy with
the print quality before you send the files.
\begin{figure}
\centering
\includegraphics[height=6.2cm]{eijkel2}
\caption{One kernel at $x_s$ (\emph{dotted kernel}) or two kernels at
$x_i$ and $x_j$ (\textit{left and right}) lead to the same summed estimate
at $x_s$. This shows a figure consisting of different types of
lines. Elements of the figure described in the caption should be set in
italics, in parentheses, as shown in this sample caption.}
\label{fig:example}
\end{figure}



\subsection*{Tables}

Las tablas llevan el captio arriba (al contrario de las figuras.

\begin{table}[h]
\caption*{Table caption}\\
\centering
\begin{tabular}{l l l}
\hline
\textbf{Treatments} & \textbf{Response 1} & \textbf{Response 2}\\
\hline
Treatment 1 & 0.0003262 & 0.562 \\
Treatment 2 & 0.0015681 & 0.910 \\
Treatment 3 & 0.0009271 & 0.296 \\
\hline
\end{tabular}
%\caption{Table caption}
\end{table}



Please define figures (and tables) as floating objects. Please avoid
using optional location parameters like ``\verb+[h]+" for ``here".

\paragraph{Remark 1.}

In the printed volumes, illustrations are generally black and white
(halftones), and only in exceptional cases, and if the author is
prepared to cover the extra cost for color reproduction, are colored
pictures accepted. Colored pictures are welcome in the electronic
version free of charge. If you send colored figures that are to be
printed in black and white, please make sure that they really are
legible in black and white. Some colors as well as the contrast of
converted colors show up very poorly when printed in black and white.

\subsection*{Formulas y Ecuaciones}

Displayed equations or formulas are centered and set on a separate
line (with an extra line or halfline space above and below). Displayed
expressions should be numbered for reference. The numbers should be
consecutive within each section or within the contribution,
with numbers enclosed in parentheses and set on the right margin --
which is the default if you use the \emph{equation} environment, e.g.,
\begin{equation}
  \psi (u) = \int_{o}^{T} \left[\frac{1}{2}
  \left(\Lambda_{o}^{-1} u,u\right) + N^{\ast} (-u)\right] dt \;  .
\end{equation}

Equations should be punctuated in the same way as ordinary
text but with a small space before the end punctuation mark.

Se recomienda no utilizar el editor de fórmulad de word que no sean el de office 365.


\subsection*{Notas a pie de página}

The superscript numeral used to refer to a footnote appears in the text
either directly after the word to be discussed or -- in relation to a
phrase or a sentence -- following the punctuation sign (comma,
semicolon, or period). Footnotes should appear at the bottom of
the
normal text area, with a line of about 2~cm set
immediately above them.\footnote{The footnote numeral is set flush left
and the text follows with the usual word spacing.}

\subsection*{Código de Programa}

Program listings or program commands in the text are normally set in
typewriter font, e.g., CMTT10 or Courier.

\medskip

\noindent
{\it Example of a Computer Program}
\begin{verbatim}
program Inflation (Output)
  {Assuming annual inflation rates of 7%, 8%, and 10%,...
   years};
   const
     MaxYears = 10;
   var
     Year: 0..MaxYears;
     Factor1, Factor2, Factor3: Real;
   begin
     Year := 0;
     Factor1 := 1.0; Factor2 := 1.0; Factor3 := 1.0;
     WriteLn('Year  7% 8% 10%'); WriteLn;
     repeat
       Year := Year + 1;
       Factor1 := Factor1 * 1.07;
       Factor2 := Factor2 * 1.08;
       Factor3 := Factor3 * 1.10;
       WriteLn(Year:5,Factor1:7:3,Factor2:7:3,Factor3:7:3)
     until Year = MaxYears
end.
\end{verbatim}
%
\noindent
{\small (Example from Jensen K., Wirth N. (1991) Pascal user manual and
report. Springer, New York)}

\subsection*{Citations y Bibliografía}

For citations in the text please use
square brackets and consecutive numbers: \cite{jour}, \cite{lncschap},
\cite{proceeding1} -- provided automatically
by \LaTeX 's \verb|\cite| \dots\verb|\bibitem| mechanism.

\subsection*{Page Numbering and Running Heads}

There is no need to include page numbers. If your paper title is too
long to serve as a running head, it will be shortened. Your suggestion
as to how to shorten it would be most welcome.

Nunca numerar los capítulos por separado. La numeración final es tarea del documento maestro que invoca a cada capítulo y organiza todos los índices.



%----------------------------------------------------------------------------------------
%	SITUACIÓN PROBLEMA
%----------------------------------------------------------------------------------------

\chapter{SITUACIÓN PROBLEMA}

\section{Identificación del problema}

\section{Formulación del problema}

\subsection{General}

\subsection{Específicos}

\section{Objetivos de la investigación}

\subsection{General}

\subsection{Específicos}

\section{Justificación}

\section{Importancia}


%----------------------------------------------------------------------------------------
%	MARCO TEORICO
%----------------------------------------------------------------------------------------

\chapter{MARCO TEÓRICO}

\section{Antecedentes de estudio}

\section{Estado del Arte}

\section{Revisión Bibliográfica}

\section{Patentes}

\section{Prinrcipales autores}

\subsection{Autores del main stream}

\newpage

\subsection{Autores disidentes}

\subsection{Escuelas antagonistas} 

\section{Crítica constructiva a las soluciones expuestas en la bibliografía}

  
\section{Hipótesis preliminar (pregunta de investigación)}

\section{Marco conceptual}

\subsection{Materiales 1}

\subsection{Datasets - Materiales 2}

\subsection{Recursos Geomáticos Materiales 3}

\subsection{Bases de Datos Materiales 4}

\subsection{Metodologías 1}

\subsection{Metodologías 2}

%----------------------------------------------------------------------------------------
%	VARIABLES E HIPÓTESIS
%----------------------------------------------------------------------------------------

\chapter{VARIABLES E HIPÓTESIS}

\section{Variables de la investigación}

\subsection{Variable dependiente}

\subsection{Variable independiente}

\section{Operacionalización de variables}

\section{Hipótesis de la investigación}

\subsection{General}

\subsection{Específicos}



%----------------------------------------------------------------------------------------
%	METODOLOGÍA
%----------------------------------------------------------------------------------------
\chapter{METODOLOGÍA}

\section{Tipo de investigación}

\section{Diseño de la investigación}

%---------------------------------------------------------------------------------------
% Eliminar los métodos que no correspondan
\subsection{Método Fáctico}
\subsection{Método Hipotético Deductivo}
\subsection{Método Empírico}
\subsection{Método Simulación}
\subsection{Método Escenarios Históricos}
%
%---------------------------------------------------------------------------------------


\section{Población y muestra}

\newpage 

\section{Técnicas e instrumentos de recolección de datos}

\subsection{Técnicas de muestreo}

\subsection{Recopilación de datos}

\subsection{Procesamiento}

\section{Procedimientos de recolección de datos}

\section{Procesamiento estadístico y análisis de datos}

\section{Resumen de resultados}

%----------------------------------------------------------------------------------------
%	RESULTADOS
%----------------------------------------------------------------------------------------

\chapter{RESULTADOS}

%----------------------------------------------------------------------------------------
%	DISCUSIÓN DE RESULTADOS
%----------------------------------------------------------------------------------------

\section{DISCUSIÓN DE RESULTADOS}


\section{Constratación de hipótesis con los resultados}

\section{Contrastación de resultados con otros estudios similares}

\section{Interpretación de los resultados}

\section{Alcance de la veracidad de las afirmaciones}

\section{Sesgo o riesgos en los datos de partida}

\section{Recomendaciones al lector}

%----------------------------------------------------------------------------------------
%	CONCLUSIONES
%----------------------------------------------------------------------------------------

\chapter{ CONCLUSIONES}

\section{Conclusiones}
Manejo de bibliografia  cob bibtex\\
Según lo expuesto por \cite{gheorghe_adapting_2001} \\
ver bibliografia al final.

\section{Aportes}

\section{Líneas de Investigación Futuras}

\section{Posibilidades de transferencia al ecosistema local}

\section{Hechos no demostrados}


\section{Mapa Mental de la Tesis}


%----------------------------------------------------------------------------------------
%	RECOMENDACIONES
%----------------------------------------------------------------------------------------

\chapter{\hspace{0.23cm}RECOMENDACIONES}

\section{Recomendaciones}

%----------------------------------------------------------------------------------------
%	BIBLIOGRAFIA
%----------------------------------------------------------------------------------------

\bibliographystyle{acm} % estilo de la bibliografia
\renewcommand*{\bibname}{REFERENCIAS}
\bibliography{bibliografia.bib} % nombre del archivo .bib
\addcontentsline{toc}{chapter}{REFERENCIAS}
\newpage


%----------------------------------------------------------------------------------------
%	ANEXOS
%----------------------------------------------------------------------------------------

\appendix
\clearpage
\addappheadtotoc
\appendixpage

\chapter{Matriz de consistencia}

\chapter{Anexo para respaldo de la investigación}

\end{document} 


